Author's response to reviews

Title: Evaluation of the Rapid Scale-up of Collaborative TB/HIV Activities in TB Facilities in Rwanda, 2005-2009

Authors:

Eric S Pevzner (ecp9@cdc.gov)
Greet Vandebriel (gv2124@columbia.edu)
David W Lowrance (LowranceD@rw.cdc.gov)
Michel Gasana (michelgasana@yahoo.fr)
Alyssa Finlay (avf0@cdc.gov)

Version: 10 Date: 6 July 2011

Author's response to reviews:

I made multiple changes to correct spelling and improve grammar throughout the document.

Also, uploaded figure 2 again because the 1 in "100%" was partially cutoff along the vertical axis.